\documentclass[10pt,a4paper]{article} \usepackage{amssymb} \usepackage{amscd} \usepackage{latexsym}
\usepackage{amsmath} \usepackage{amsfonts} \usepackage{amscd} \usepackage{xcolor} \usepackage{ulem}

\begin{document}
\begin{thebibliography}{99}

\bibitem{BS} G. Braun and L. Str\"ungmann, {\it The independence of the notions of Hopfian and co-Hopfian Abelian $p$-groups}, Proc. Am. Math. Soc. {\bf 143}(8) (2015), 3331--3341.

\bibitem{CDK} A. R. Chekhlov, P. V. Danchev and P. W. Keef, {\it Two generalizations of co-Hopfian Abelian groups}, Sib. Math. J. {\bf 67}(1) (2026).

\bibitem{D} G. D'Este, {\it The $\oplus_c$-topology on abelian $p$-groups}, Ann. Scuola Norm. Sup. Pisa, Serie
    IV {\bf 7}(2) (1980), 241--256.

\bibitem{F1} L. Fuchs, Infinite Abelian Groups, vols. {\bf I} and {\bf II}, Acad. Press, New York and London,
    1970 and 1973.

\bibitem{F2} L. Fuchs, Abelian Groups, Springer, Switzerland (2015).

\bibitem{GG} B. Goldsmith and K. Gong, {\it A note on Hopfian and co-Hopfian Abelian groups}, Contemp. Math.
    {\bf 576} (2012), 129--136.

\bibitem{IK} J. Irwin and P. Keef, {\it Primary abelian groups and direct sums of cyclics}, J. Algebra {\bf
    159}(2) (1993), 387--399.

\bibitem{Ka} E. V. Kaigorodov, {\it Hopfian completely decomposable torsion-free groups}, Vest. Tomsk University (Math. and Mech.) {\bf 24} (2013), 24--28.

\bibitem{KS} P. Keef and L. Salce, {\it Small pairs of abelian $p$-groups}, J. Pure Appl. Algebra {\bf 227}(4)
    (2023).

\bibitem{Me} C. Megibben, {\it Large subgroups and small homomorphisms}, Michigan Math. J. {\bf 13} (1966),
    153--160.

\end{thebibliography}


\newtheorem{theorem}{Theorem}[section] \newtheorem{remark}[theorem]{Remark} \newtheorem{mtheorem}[theorem]{Main
Theorem} \newtheorem{bbtheo}[theorem]{The Strong Black Box} \newtheorem{observation}[theorem]{Observation}
\newtheorem{proposition}[theorem]{Proposition} \newtheorem{lemma}[theorem]{Lemma}
\newtheorem{testlemma}[theorem]{Test Lemma} \newtheorem{mlemma}[theorem]{Main Lemma}
\newtheorem{note}[theorem]{{\bf Note}} \newtheorem{steplemma}[theorem]{Step Lemma}
\newtheorem{corollary}[theorem]{Corollary} \newtheorem{notation}[theorem]{Notation}
\newtheorem{example}[theorem]{Example} \newtheorem{definition}[theorem]{Definition}

\renewcommand{\labelenumi}{(\roman{enumi})} \def\Pf{\smallskip\goodbreak{\sl Proof.}}

\def\Fin{\mathop{\rm Fin}\nolimits} \def\br{\mathop{\rm br}\nolimits} \def\fin{\mathop{\rm fin}\nolimits}
\def\Ann{\mathop{\rm Ann}\nolimits} \def\Aut{\mathop{\rm Aut}\nolimits} \def\End{\mathop{\rm End}\nolimits}
\def\Tor{\mathop{\rm Tor}\nolimits} \def\bfb{\mathop{\rm\bf b}\nolimits} \def\bfi{\mathop{\rm\bf i}\nolimits}
\def\bfj{\mathop{\rm\bf j}\nolimits} \def\df{{\rm df}} \def\bfk{\mathop{\rm\bf k}\nolimits}
\def\bEnd{\mathop{\rm\bf End}\nolimits} \def\iso{\mathop{\rm Iso}\nolimits} \def\id{\mathop{\rm id}\nolimits}
\def\Ext{\mathop{\rm Ext}\nolimits} \def\Ines{\mathop{\rm Ines}\nolimits} \def\Hom{\mathop{\rm Hom}\nolimits}
\def\bHom{\mathop{\rm\bf Hom}\nolimits} \def\Rk{ R_\k-\mathop{\bf Mod}} \def\Rn{ R_n-\mathop{\bf Mod}}
\def\map{\mathop{\rm map}\nolimits} \def\cf{\mathop{\rm cf}\nolimits} \def\top{\mathop{\rm top}\nolimits}
\def\Ker{\mathop{\rm Ker}\nolimits} \def\Bext{\mathop{\rm Bext}\nolimits} \def\Br{\mathop{\rm Br}\nolimits}
\def\dom{\mathop{\rm Dom}\nolimits} \def\min{\mathop{\rm min}\nolimits} \def\im{\mathop{\rm Im}\nolimits}
\def\max{\mathop{\rm max}\nolimits} \def\rk{\mathop{\rm rk}} \def\Diam{\diamondsuit} \def\Z{{\mathbb Z}}
\def\Q{{\mathbb Q}} \def\N{{\mathbb N}} \def\bQ{{\bf Q}} \def\bF{{\bf F}} \def\bX{{\bf X}} \def\bY{{\bf Y}}
\def\bHom{{\bf Hom}} \def\bEnd{{\bf End}} \def\bS{{\mathbb S}} \def\AA{{\cal A}} \def\BB{{\cal B}} \def\CC{{\cal
C}} \def\DD{{\cal D}} \def\TT{{\cal T}} \def\FF{{\cal F}} \def\GG{{\cal G}} \def\PP{{\cal P}} \def\SS{{\cal S}}
\def\XX{{\cal X}} \def\YY{{\cal Y}} \def\fS{{\mathfrak S}} \def\fH{{\mathfrak H}} \def\fU{{\mathfrak U}}
\def\fW{{\mathfrak W}} \def\fK{{\mathfrak K}} \def\PT{{\mathfrak{PT}}} \def\T{{\mathfrak{T}}} \def\fX{{\mathfrak
X}} \def\fP{{\mathfrak P}} \def\X{{\mathfrak X}} \def\Y{{\mathfrak Y}} \def\F{{\mathfrak F}} \def\C{{\mathfrak
C}} \def\B{{\mathfrak B}} \def\J{{\mathfrak J}} \def\fN{{\mathfrak N}} \def\fM{{\mathfrak M}} \def\Fk{{\F_\k}}
\def\bar{\overline } \def\Bbar{\bar B} \def\Cbar{\bar C} \def\Pbar{\bar P} \def\etabar{\bar \eta} \def\Tbar{\bar
T} \def\fbar{\bar f} \def\nubar{\bar \nu} \def\rhobar{\bar \rho} \def\Abar{\bar A} \def\a{\alpha} \def\b{\beta}
\def\g{\gamma} \def\w{\omega} \def\e{\varepsilon} \def\o{\omega} \def\va{\varphi} \def\k{\kappa} \def\m{\mu}
\def\n{\nu} \def\r{\rho} \def\f{\phi} \def\hv{\widehat\v} \def\hF{\widehat F} \def\v{\varphi} \def\s{\sigma}
\def\l{\lambda} \def\lo{\lambda^{\aln}} \def\d{\delta} \def\z{\zeta} \def\th{\theta} \def\t{\tau}
\def\ale{\aleph_1} \def\aln{\aleph_0} \def\Cont{2^{\aln}} \def\nld{{}^{ n \downarrow }\l} \def\n+1d{{}^{ n+1
\downarrow }\l} \def\hsupp#1{[[\,#1\,]]} \def\size#1{\left|\,#1\,\right|} \def\Binfhat{\widehat {B_{\infty}}}
\def\Zhat{\widehat \Z} \def\Mhat{\widehat M} \def\Rhat{\widehat R} \def\Phat{\widehat P} \def\Fhat{\widehat F}
\def\fhat{\widehat f} \def\Ahat{\widehat A} \def\Chat{\widehat C} \def\Ghat{\widehat G} \def\Bhat{\widehat B}
\def\Btilde{\widetilde B} \def\Ftilde{\widetilde F} \def\restr{\mathop{\upharpoonright}} \def\to{\rightarrow}
\def\arr{\longrightarrow} \def\LA{\langle} \def\RA{\rangle} \newcommand{\norm}[1]{\text{$\parallel\! #1
\!\parallel$}} \newcommand{\supp}[1]{\text{$\left[ \, #1\, \right]$}} \def\set#1{\left\{\,#1\,\right\}}
\newcommand{\mb}{\mathbf} \newcommand{\wt}{\widetilde} \newcommand{\card}[1]{\mbox{$\left| #1 \right|$}}
\newcommand{\union}{\bigcup}%\limits} \newcommand{\inters}{\bigcap}%\limits} \newcommand{\ER}{{\rm E}}
\def\Proof{{\sl Proof.}\quad} \def\fine{\ \black\vskip.4truecm} \def\black{\ {\hbox{\vrule width 5pt height 5pt
depth 0pt}}} \def\fine{\ \black\vskip.5truecm} \long\def\alert#1{\smallskip\line{\hskip\parindent\vrule%
\vbox{\advance\hsize-2\parindent\hrule\smallskip\parindent.5\parindent%
\narrower\noindent#1\smallskip\hrule}\vrule\hfill}\smallskip}

\title{Notes on Some Versions of \\ co-Hopfian Abelian Groups}
\footnotetext{2010 AMS Subject Classification: Primary 20K10, 20K12; Secondary 20K20, 20K21, 20K30.
Key words and phrases: Abelian groups, directly finite groups, co-Hopfian groups, co-Bassian groups, generalized co-Bassian groups}

\author{Andrey R. Chekhlov
\\Faculty of Mathematics and Mechanics, Section of Algebra, \\Tomsk State University, Tomsk, Russia\\{\small
e-mails: cheklov@math.tsu.ru, a.r.che@yandex.ru}\\
Peter V. Danchev \\Institute of Mathematics and Informatics,
Section of Algebra, \\Bulgarian Academy of Sciences, Sofia, Bulgaria\\{\small e-mails: danchev@math.bas.bg,
pvdanchev@yahoo.com} \\and\\
Patrick W. Keef \\ Department of Mathematics, \\ Whitman College, Walla Walla, WA,
Unites States\\{\small e-mail: keef@whitman.edu}} \maketitle

\centerline{(To Brendan Goldsmith on his {\bf 75}th birthday)}

\begin{abstract}{We establish some additional results concerning relatively co-Hopfian Abelian groups, recently defined by us in the Siber. Math. J. (2026). Concretely, for any prime $p$, we prove that relatively co-Hopfian $p$-groups are thick as well as that separable relatively co-Hopfian $p$-groups are thin. Moreover, we show that a reduced fully-starred $p$-group is relatively co-Hopfian exactly when it is finite. Finally, we demonstrate that relatively co-Hopfian groups are closed under the formation of finitely generated extension.}
\end{abstract}

\section{Introduction and Background}

Throughout this article, all groups into consideration are {\it additively} written and {\it Abelian}. Our basic notation and terminology will standardly follow those from the books \cite{F1,F2}. As usual, for some prime integer $p$, $\Z(p^n)$ denotes the cyclic $p$-group of order $p^n$ for some $n\geq 1$, and $\Z(p^{\infty})$ stands for the quasi-cyclic divisible $p$-group.

\medskip

A classical concept of some majority in certain directions of Abelian group theory is the one of a {\it co-Hopfian}
group $G$ which states that any monomorphism (in other terms, an injective endomorphism) $\phi: G\to G$ is surjective, i.e., $\phi(G)=G$ (or, equivalently, $\phi$ is an bijective homomorphism, that is, an isomorphism). So, $\phi(G)$ is a trivial (i.e., a non-proper) direct summand of $G$. In other words, restating this in an equivalent manner, $G$ is a
group that is not isomorphic to any of its proper subgroups.

\medskip

This restatement motivated us in \cite{CDK} to formulate and study in-depth the following two notions.

\medskip

\noindent{\bf $\bullet$} We shall say that a group $G$ is {\it relatively co-Hopfian} if, for every
monomorphism, $\phi: G \to G$, the image $\phi(G)$ is essential in $G$.

\medskip

\noindent{\bf $\bullet$} We shall say that a group $G$ is {\it generalized co-Hopfian} if, whenever $\phi:G\to G$ is
a monomorphism, then $\phi(G)$ is a direct summand of $G$.

\medskip

In what follows, we are planning to discover some extra substantial results on these two classes of groups, thus significantly supplying our results from \cite{CDK}. Specifically, we are succeeding in obtaining the following achievements: for any prime $p$, relatively co-Hopfian $p$-groups are thick (see Proposition~\ref{coHopfiansarethick}). Likewise, we show that separable relatively co-Hopfian $p$-groups are thin (see Proposition~\ref{coHopfiansarethin}). Also, we prove that reduced fully-starred relatively co-Hopfian $p$-groups are finite (see Proposition~\ref{fullstar}). Besides, we establish that the property of being relatively co-Hopfian is inherited under the formation of its direct sum with an arbitrary finite generation (see Proposition~\ref{dirsumprop1}).

\section{Results and Examples}

Standardly, as noticed above, for the maximal torsion subgroup $T(G)=\oplus_p T_p(G)$ of a group $G$, the symbol $T_p(G)$ stands for its $p$-torsion component for some prime $p$.

\medskip

The next statement is somewhat helpful and close to the preceding assertion, so it is worthy of documentation
(at least that it seems to be unpublished in the existing literature).

\begin{proposition} The subgroup $p^nG$ is essential in $G$ for every integer $n>0$ if, and only if, $T_p(G)$ is divisible.
\end{proposition}

\Pf ~ {\bf Necessity.} If, in a way of contradiction, we assume that $T_p(G)\neq\{0\}$ is not divisible, then
$T_p(G)$ contains a non-zero cyclic direct summand, say $H$, of $G$ of order $k>0$, and hence $H\cap
p^{k+1}G=\{0\}$, as pursued.

\medskip

{\bf Sufficiency.} Letting $H$ be a non-zero cyclic subgroup in $G$, if $H\cap T_p(G)\neq\{0\}$, then $H\cap
p^nG\neq\{0\}$ for every integer $n>0$. Furthermore, if $H\cap T_p(G)=\{0\}$, then it is not so difficult to
verify that it is possible to choose the subgroup $B$ of $G$ such that $G=T_p(G)\oplus B$ and $H\leq B$ (see,
e.g., \cite[Theorem 21.2]{F1}). But, since $T_p(B)=\{0\}$, we deduce that $\{0\}\neq p^nH\leq p^nG$ for every
$n>0$, as asked for.
\fine

Before proceeding by proving our next result on direct sums, we need one more technicality like this.

\begin{lemma}\label{technicalessential} Let $G=A\oplus B$ with projection $\pi: G\to A$, and let $H=K\oplus N\oplus F\leq G$, where $N$ is an essential subgroup of $B$ and the subgroup $\pi(K)$ is essential in $A$. Then, $F=\{0\}$.
\end{lemma}

\Pf ~ Since $N$ is essential in $B$, we have $F\cap B=\{0\}$. We now intend to prove that $F\cap A=\{0\}$, so assume the contrary that $0\neq x\in F\cap A$. Since $\pi(K)$ is essential in $A$, there exists $n\in\mathbb{N}$ such that $nx\neq 0$ and $nx+b\in K$ for some $b\in B$. Note that $b\neq 0$, as for otherwise $0\neq nx\in F\cap K=\{0\}$. If now $0\neq mb\in N$ for $m\in\mathbb{N}$, then $$nmx=(nmx+mb)-mb\in(K\oplus N)\cap F=\{0\}.$$ Therefore, $nmx=0$ and hence $$0\neq mb=nmx+mb\in K\cap N=\{0\},$$ a contradiction. So, it must be that $F\cap A=\{0\}$.

Assume now that $F\neq\{0\}$ and $0\neq x\in F$. Then, $x=a+b$, where, in view of the already obtained above equalities $F\cap A=F\cap B=\{0\}$, we have $0\neq a\in A$, $0\neq b\in B$. Thus, there exists $m\in\mathbb{N}$ such that $ma\neq 0$ and $y:=ma+b'\in K$ for some $b'\in B$. Hence, $mx-y=mb-b'\in B$, where $mx-y\neq 0$ since $mx\neq 0$, $y\neq 0$ and $K\cap F=\{0\}$. Consequently, $b_1:=mb-b'\neq 0$, where $b_1\in B$. Observe also that there exists $n\in\mathbb{N}$ such that $0\neq nb_1\in N$. Thus, $$0\neq nb_1=nmx-ny\in (F\oplus K)\cap N=\{0\}.$$ This contradiction gives that $F=\{0\}$, as claimed.
\fine

It was proved in \cite[Lemma 2.1]{CDK} that a direct summand of a relatively co-Hopfian group is too relatively co-Hopfian. Moreover, it was proven in \cite[Proposition 2.6]{CDK} that the direct sum of two relatively co-Hopfian groups, one of which is fully invariant in this sum, is again relatively co-Hopfian.

\medskip

We now have enough instruments to attack the truthfulness of this claim by using a more transparent approach.

\begin{proposition}\label{fullyinv2} Let $G=A\oplus B$, where $B$ is fully invariant in $G$ and both $A$, $B$
are relatively co-Hopfian groups. Then, $G$ also is a relatively co-Hopfian group.
\end{proposition}

\Pf ~ Write $$H=K\oplus N\oplus F\leq G,$$ where $K\cong A$ and $N\cong B$. Since $B$ is fully invariant in $G$, we
get $N\leq B$, and since $N\cong B$, we have that $N$ is essential in $B$ by referring to the corresponding definition listed above. So, $K\cap B=\{0\}$ whence, if $\pi: G\to A$ is a projection, then having in mind the isomorphisms $$\pi(K)\cong K\cong A,$$ one verifies that the subgroup $\pi(K)$ is essential in $A$ by the same definition. Consequently, Lemma~\ref{technicalessential} tells us that $F=\{0\}$. So, the group $G$ is really relatively co-Hopfian, as asserted.
\fine

Recall that a $p$-group $G$ is said to be {\it thick}, if whenever $B$ is a direct sum of cyclic $p$-groups and
$\phi: G\to B$ is a homomorphism, then $\phi$ is {\it small}, i.e., the kernel of $\phi$ is a {\it large}
subgroup of $G$ (for a more detailed information, we refer to \cite[Section 67]{F1}).

\medskip

To use this notion in the present context, we recall the following helpful result:

\medskip

\noindent{\bf $\bullet$} (\cite[Corollary~18(a)]{IK}) {\it If $G$ is a $p$-group that is not thick, then there is a $p$-group $H$ with a direct summand that is an unbounded direct sum of cyclic groups such that $G$ and $H$ embed in each other.}

\medskip

We, thereby, come to the following more concrete statement.

\begin{proposition}\label{coHopfiansarethick} If the $p$-group $G$ is relatively co-Hopfian, then it is thick.
\end{proposition}

\Pf ~ Suppose the contrary that $G$ is {\it not} thick and $H=A\oplus B$, where $B$ is an unbounded direct sum of cyclic groups such that $G$ and $H$ embed in each other. Since $B$, and hence $H$, is {\it not} relatively co-Hopfian, it follows that $H$ embeds into itself as a non-essential subgroup. Therefore, $G$ also embeds in $H$ as a
non-essential subgroup. But since $H$ embeds in $G$, one has that $G$ must embed in itself as a non-essential subgroup. Consequently, $G$ is {\it not} relatively co-Hopfian, as expected.
\fine

Note that an unbounded torsion-complete $p$-group will be thick, but not co-Hopfian, so the converse to the last result does {\it not} generally hold. It, however, does have in what follows an interesting consequence.

\medskip

Recall that the so-called {\it $\oplus_c$-topology} uses the subgroups $X\leq G$ such that $G/X$ is a direct sum of cyclic groups as a neighborhood base of $\{0\}$ (see, for a more account, \cite{D}).

\begin{corollary}\label{opluscomplete} Suppose $G$ is a relatively co-Hopfian $p$-group. Then, $G$ is complete
in its $\oplus_c$-topology if, and only if, it is finite.
\end{corollary}

\Pf ~ Sufficiency being pretty obvious, we now focus on the necessity. To that end, suppose now that $G$ is complete in its $\oplus_c$-topology; in particular, $G$ must be separable. Since $G$ is thick, one knows that its $\oplus_c$-completion agrees with its torsion-completion, say $\overline G$ (see cf. \cite[Proposition~1.1]{D}). So, \cite[Lemma 2.21]{CDK} works successfully to derive that $G\cong \overline G$ must be finite, as stated.
\fine

We now mention a specific case of the last above result. In \cite{F2}, L. Fuchs designated the smallest class of Abelian $p$-groups containing the cyclic groups that is closed with respect to direct sums, direct summands, and the torsion subgroups of direct products over non-measurable index sets as the {\it Keef class}, denoting it by $K_p$. Clearly, $K_p$ also contains both the direct sums of cyclic $p$-groups and the torsion complete
$p$-groups. It is principally known that the elements of $K_p$ are all complete in their $\oplus_c$-topologies.

\medskip

Therefore, the following statement is an immediate consequence of Corollary~\ref{opluscomplete}.

\begin{corollary} A group $G\in K_p$ is relatively co-Hopfian if, and only if, it is finite.
\end{corollary}

We now consider a property that is, in some sense, dual to that of being thick. The $p$-group $G$ is said to be {\it thin} if, for every torsion-complete $p$-group $\overline B$, any homomorphism $\phi: \overline{B}\to G$ must be small. In this vein, Megibben showed that, if $G$ is separable, then $G$ is thin if, and only if, it does not have a subgroup that is isomorphic to an unbounded torsion-complete $p$-group (\cite[Theorem~3.1]{Me}). Thus, the following relates this to our inquiry.

\begin{proposition}\label{coHopfiansarethin} If $G$ is a separable $p$-group that is relatively co-Hopfian, then $G$ is thin.
\end{proposition}

\Pf ~ We suppose the contrary that $G$ is a separable $p$-group that is not thin, and show that $G$ is also not
relatively co-Hopfian. By the aforementioned result from \cite{Me}, $G$ must have a subgroup $H\cong \overline B$, where $B$ is an unbounded direct sum of cyclic groups. Utilizing now a familiar argument, $\overline B$ has an unbounded proper direct summand, and replacing $H$ with this summand, there is no loss of generality in assuming that $H$ is not essential in $G$.

Note that since the group $G$ is semi-standard (that is, all its Ulm-Kaplansky invariants defined over the set of finite ordinals always have finite value), if $A$ is a basic subgroup of $G$, then $A$ is also semi-standard, and hence the direct sum of a countable number of cyclic direct summands. By embedding cyclic direct summands of $A$ into larger cyclic direct summands of $B$, there is a monomorphism $\gamma:A\to B$ which extends to a monomorphism $\overline A\to \overline B$. Since $G$ embeds in $\overline A$ which embeds in $\overline B$ which embeds as a non-essential subgroup of $G$ again, we can conclude that $G$ is not relatively co-Hopfian, as stated.
\fine

It follows from a combination of Propositions~\ref{coHopfiansarethick} and \ref{coHopfiansarethin} that a
separable relatively co-Hopfian $p$-group is {\it thick-thin}, a class of groups that received considerable
attention in \cite{KS}.

\medskip

Recall that the reduced $p$-group $G$ is said to be {\it fully starred} if every subgroup of $G$ has the same
cardinality as one of its basic subgroups. For example, if $H$ and $K$ are reduced $p$-groups, it is known that the {\it torsion functor} $\Tor (H,K)$ between $H$ and $K$ (and all of its subgroups) will be fully starred (see, for example, cf. \cite[Chapter X]{F1}). In particular, this implies that, for any ordinal $\alpha$, any subgroup of a $p^\alpha$-pure projective $p$-group will be fully starred. In addition, a countable reduced $p$-group is fully starred.

\medskip

We can now record the following.

\begin{proposition}\label{fullstar} If $G$ is a reduced fully-starred $p$-group, then $G$ is relatively co-Hopfian if, and only if, it is finite.
\end{proposition}

\Pf ~ Certainly, if $G$ is finite, then it is co-Hopfian and thus relatively co-Hopfian.

So, assume $G$ is an infinite reduced and fully-starred relatively co-Hopfian $p$-group. Since $G$ must be semi-standard, if it is bounded, then it must be finite. Reciprocally, if it is assumed on the contrary to be unbounded, then it must have a countably infinite basic subgroup, so that $G$ is also simultaneously countably infinite and unbounded. Therefore, $G/p^\omega G$ is too countably infinite and unbounded, and hence an unbounded direct sum of cyclic groups being separable (see, for instance, \cite[Chapter III, Section 17]{F1} or
\cite[Chapter 3]{F2}). But, an appeal to \cite[Proposition 2.18]{CDK} reaches that the group $G$ is co-Hopfian, and thus \cite{BS} enables us that the factor-group $G/p^\omega G$ is co-Hopfian as well, and so Proposition~\ref{coHopfiansarethick} allows us to infer that the group $G$, and hence the quotient $G/p^\omega G$, must be thick, giving the desired contradiction.
\fine

The following result on co-Hopficity is, principally, well-documented (see, e.g., \cite{GG}).

\begin{corollary} A countable reduced $p$-group is relatively co-Hopfian if, and only if, it is finite.
\end{corollary}

Consulting with \cite[Proposition 2.23]{CDK}, we may derive the following helpful consequence.

\begin{corollary}\label{BcoH} Suppose $G$ is a group with torsion $T$. If $T$ is co-Hopfian and $G/T$ has finite rank, then $G$ is relatively co-Hopfian.
\end{corollary}

The following comments could somewhat be useful. They illustrate that the restrictions used by us in our affirmations are the most minimal possible.  

\begin{remark} It is worthwhile noticing that, exploiting an analogous idea, the cited assertion from \cite{CDK} can be extended thus: {\it If $F$ is a fully invariant subgroup of a group $G$ such that $F$ is co-Hopfian and $G/F$ is relatively co-Hopfian, then $G$ itself is relatively co-Hopfian}.

\medskip

It is also worthy of noticing that, for every prime $p$, let $Q_p\subseteq \Q$ be the group consisting of the fractions of the form $a/p^k$ such that $a\in \Z$, $k\in \N$. In particular, for all $p\in \cal P$, the group $Q_p$ has type greater than $\Z$ and there are an infinite number of such $Q_p$. Considering now the group $$G:=\Z\oplus(\bigoplus_{p\in \cal P} Q_p),$$ we, however, discover that this group $G$ pretty clearly is relatively co-Hopfian. This expectedly satisfies \cite[Proposition 2.26]{CDK} in which proposition we require the condition for non-existence of an infinite subset $\{i_k ~ : ~ k\in \N\}$ such that $\t_k < \t_{k+1}$ for each $k\in \N$. This is, however, in a sharp contrast with the requirement that "each $t\in\Omega$ has only finite type greater than $t$" and, that is why, a claim of the sort "{\it The completely decomposable torsion-free group $G$ for which $G=\bigoplus_{t\in\Omega}G_t$, where all direct summands $G_t$ are homogeneous components of $G$ and $\Omega$ is a set of same type direct summands of $G$ of rank~1, is relatively co-Hopfian if, and only if, each $G_t$ has finite rank and each $t\in\Omega$ has only finite type greater than $t$}" cannot happen as the constructed above group $G$ unambiguously shows (compare with \cite{Ka} as well).
\end{remark}

\medskip

We are now concerned with the inheritance of the direct sum property via the relative co-Hopficity, intending to
confirm the following two statements, the first of which generalizes significantly the corresponding well-known result for
co-Hopficity.

\begin{proposition}\label{dirsumprop1} If $G$ is a relatively co-Hopfian group and $F$ is a finitely generated
group, then $G\oplus F$ too is a relatively co-Hopfian group.
\end{proposition}

\Pf ~ One sees that it is enough to prove the assertion when $F=\langle a\rangle$ is a cyclic group of either
infinite order or an order $p^n$ for some prime number $p$ and an integer $n>0$.

To this target, suppose that $A=G\oplus\langle a\rangle$ and $\pi: A\to G$ is the corresponding projection.
Assume that $H=C\oplus \langle b\rangle\oplus X\lneqq A$, where $C\cong G$, ${\rm order}(b)={\rm order}(a)$ and $X\neq\{0\}$.

Let ${\rm order}(a)=p^n$. Consider two basic cases as follows:

\medskip

\noindent{\bf Case 1:} Write $b=g+a$ for some $g\in G$ (accurate to certain integer multiple, mutually simple
with $p$). We have $A=G\oplus\langle a+g\rangle$ and so $H=(H\cap G)\oplus\langle a+g\rangle$. Here, $H\cap
G\cong C\oplus X$ and $H\cap G\lneqq G$ since relatively co-Hopfian groups do not have proper direct summands
isomorphic to itself. However, as $C\cong G$ and $X\neq\{0\}$, which is impossible for a relatively co-Hopfian
group.

\medskip

\noindent{\bf Case 2:} Write $b=g+p^la$, where $0<l\leq n$. Thus, ${\rm order}(b)={\rm order}(g)$. Since
$p^{n-l}b=p^{n-l}g\neq 0$, it must be that $\langle g\rangle\cap(C\oplus X)=\{0\}$. We, thus, arrive at the
subgroup $H'=C\oplus\langle g\rangle\oplus X\leq A$, which obviously is a proper direct decomposition.

\medskip

If $C\cap\langle a\rangle=\{0\}$, then $\pi(C)\oplus\pi(\langle g\rangle)\lneqq G$, where $\pi(C)\cong C\cong G$ and $\pi(\langle g\rangle)=\langle g\rangle$, which does not hold for a relatively co-Hopfian group.

Assume now that $C\cap\langle a\rangle\neq\{0\}$, so that $c=p^ta\in C$ for some $0\leq t<n$. Therefore,
$C\cap\langle a+g\rangle=\{0\}$. In fact, if $c_1=p^s(a+g)\in C$ for some $0\leq s<n$, then, for $m=\max\{t,s\}$, we have $$0\neq p^{m-s}c_1-p^ma=p^mg\in\langle g\rangle\cap C,$$ a contradiction, as expected.

Next, if $\pi': A\to G$ is the corresponding projection of a group $A=G\oplus\langle a+g\rangle$ on $G$, then $$\pi'(C\oplus\langle g\rangle)= \pi'(C)\oplus\langle g\rangle\lneqq G,$$ where $\pi'(C)\cong G$, which is manifestly wrong for a relatively co-Hopfian group.

Finally, let ${\rm order}(a)=\infty$. Assume $b=g+ka$ for some integer $k\neq 0$. So, $$H=(H\cap G)\oplus\langle g+ka\rangle\leq A'=G\oplus \langle g+ka\rangle,$$ where $H\cap G\cong C\oplus X$ and $H\cap G\lneqq G$, which is false for a relatively co-Hopfian group.

Let us now we have $b\in G$. If $C\cap\langle a\rangle=\{0\}$, then $\pi(C)\oplus \langle b\rangle\lneqq G$,
where $\pi(C)\cong G$, which is untrue for a relatively co-Hopfian group. But, if $C\cap\langle
a\rangle\neq\{0\}$, then $c=ka\in C$ for some integer $k\neq 0$. Thus, for $A=G\oplus\langle a+b\rangle$, we
obtain $C\cap\langle a+b\rangle=\{0\}$, because if $c_1=k(a+b)$, then $$nc_1-nka=nkb\in \langle b\rangle\cap
C=\{0\},$$ a contradiction. So, if $\pi'': A\to G$ is a projection of $A=G\oplus\langle a+b\rangle$ on $G$, then argued as above $\pi''(C)\oplus\langle b\rangle\lneqq G$ and $\pi''(C)\cong G$, which is manifestly {\it not} fulfilled for a relatively co-Hopfian group.
\fine

Recall that a groups is termed {\it finitely co-generated}, provided that its reduced part and divisible part have a respective finite generating set.

\medskip

As a valuable consequence, we have:

\begin{corollary} If $G$ is a relatively co-Hopfian group and $F$ is a finitely co-generated group, then
$G\oplus F$ too is a relatively co-Hopfian group.
\end{corollary}

\Pf ~ Since any finitely co-generated group is known to be a finite direct sum of co-cyclic groups (i.e., a direct sum of a finite number of cyclic and divisible groups), then the result follows from a combination between Proposition~\ref{dirsumprop1} and Proposition~\ref{fullyinv2}, equipped with the construction of a divisible relatively co-Hopfian group given in \cite{CDK}.
\fine

Further, one may expect that an analogous claim to that in Proposition~\ref{dirsumprop1} will hold for
generalized co-Hopfian groups, namely that if $G$ is a generalized co-Hopfian group and $F$ is a finitely
generated group, then $G\oplus F$ is too a generalized co-Hopfian group, but unfortunately this is {\it not}
manifestly invalid. In fact, as shown in \cite{CDK}, any torsion-free generalized co-Hopfian group is divisible, and
even if $F$ is taken to be torsion (and hence finite), \cite[Theorem 2.32]{CDK} will teach us that $f_k(F)=0$ for all $k<n$, which means that $F$ is semi-standard, that is manifestly untrue.

\section{Concluding Discussion}

In order to provide the interested reader with some more useful information, we intend here to comment some things in other light.

\medskip

They are closely connected to the procedure described in \cite[Section 4]{BS}, where it is demonstrated that if a $p$-group $G$ is co-Hopfian, then so does its first Ulm factor $G/p^{\omega}G$, as well as some other important things under the presence of the Martin's axiom (MA). We thus arrive at the following two challenging questions, the first one of which is pertained to \cite[Proposition 2.4]{CDK} and the second one to \cite[Theorem 2.32]{CDK}, but both of them are results in (ZFC) without any extra set-theoretic assumptions.

\medskip

\noindent{\bf Problem 1.} If a $p$-group $G$ is directly finite, is its first Ulm factor $G/p^{\omega}G$ also
directly finite?

\medskip

\noindent{\bf Solution.} The question follows directly from the cited above Proposition 2.4 from \cite{CDK}, because it must be that $$f_n(G/p^{\omega}G)=f_n(G)$$ for $n<\omega$ and $$f_{\alpha}(G/p^{\omega}G)=0$$ for $\alpha\geq\omega$ (see \cite[Chapter XII, Section 77]{F1}). This terminates our proof.

\medskip

Note that, if a $p$-group $G$ is relatively co-Hopfian, then its first Ulm factor $G/p^{\omega}G$ is also
relatively co-Hopfian. In fact, just a simple combination of the conclusions in \cite[Section 4]{BS} and
\cite[Proposition 2.19]{CDK} works to get the claim.

\medskip

\noindent{\bf Problem 2.} If a $p$-group $G$ is generalized co-Hopfian, is its first Ulm factor $G/p^{\omega}G$ also generalized co-Hopfian?

\medskip

\noindent{\bf Solution.} Thankfully, \cite[Theorem 2.32]{CDK} yields that $G=A\oplus C$, where $C$ is a
$p^n$-bounded group and $A$ is a co-Hopfian group. So, one writes that $$G/p^{\omega}G\cong
(A/p^{\omega}A)\oplus C.$$

Furthermore, since by what we have commented above $A/p^{\omega}A$ is co-Hopfian, and $$f_m(A/p^{\omega}A)=f_m(A)$$ for every $m<\omega$ (see \cite[Chapter XII, Section 77]{F1}), again the same \cite[Theorem 2.32]{CDK} is workable to get that $G/p^{\omega}$ is generalized co-Hopfian, as pursued. The proof is over.

\medskip

So, we come to our final question which settling is unknown to us yet.

\medskip

\noindent{\bf Problem 3.} Find suitable conditions on the first Ulm subgroup $p^{\omega} G$ of an arbitrary $p$-group $G$ under which, if the first Ulm factor $G/p^{\omega}G$ is directly finite (resp., generalized co-Hopfian), then so does the former group $G$.

\medskip

\noindent{\bf Funding:} The work of the first-named author, A.R. Chekhlov, is supported by the Ministry of
Science and Higher Education of Russia (agreement No. 075-02-2024-1437).

\medskip

\noindent{\bf Acknowledgement.} The authors would like to express their deep thankfulness to the anonymous expert referee for the numerous constructive suggestions on the whole presentation of our work given in the report.

\vskip1.0pc


\end{document}











